Author's response to reviews

Title: Rationale and design of WEBCARE: A randomized, controlled, web-based behavioral intervention trial in cardioverter-defibrillator patients to reduce anxiety and device concerns and enhance quality of life

Authors:

- Susanne S Pedersen (s.s.pedersen@uvt.nl)
- Viola Spek (v.r.m.spek@uvt.nl)
- Dominic AMJ Theuns (d.theuns@erasusmc.nl)
- Marco Alings (marco@alings.org)
- Pepijn van der Voort (pepijn.vd.voort@catharina-ziekenhuis.nl)
- Luc Jordaens (l.jordaens@erasusmc.nl)
- Pim Cuijpers (P.Cuijpers@psy.vu.nl)
- Johan Denollet (denollet@uvt.nl)
- Krista C van den Broek (C.C.L.T.vdnBroek@uvt.nl)

Version: 5 Date: 22 December 2009

Author's response to reviews:

I have added initials to two of the authors